Author's response to reviews

Title: Altered mental status, an unusual manifestation of early disseminated Lyme disease: A case report

Authors:

Shiven B Chabria (shivenchabria@yahoo.com)
Jock Lawrason (jlawrason@wtbyhosp.org)

Version: 3 Date: 2 August 2007

Author's response to reviews:

Dear Sirs,

Please find my modified article. Following modifications have been made per your communications:

Title Page - requested changes have been made to the author list, affiliations and email addresses
Title - has been reformatted with the words: a case report at the very end
Qualifications have been removed
Affiliations - have been reformatted as requested
References - superscript numbers have been replaced with regular ones. The full stop follows the reference.
Case report has been changed to case presentation
Authors contributions - have been duly updated